Author's response to reviews

Title: Chylous ascites associated with chylthorax; a rare sequela of penetrating abdominal trauma: a case report.

Authors:

Joseph M Plummer (joseph_plummer@yahoo.com)
Michael E McFarlane (michael.mcfarlane@uwimona.edu.jm)
Archibald H McDonald (archibald.mcdonald@uwimona.edu.jm)

Version: 7 Date: 30 October 2007

Author's response to reviews:

The Editor,
Journal of Medical Case Reports

Dear Sirs,

Re: Manuscript ID 1184023106147998

The above manuscript is hereby re-submitted with minor English language corrections accepted as suggested by the Editorial board. These minor changes have been tracked in the main document.

Thank you for your patience and kindness throughout this submission process.

Sincerely,

Joseph Plummer.